Author's response to reviews

Title: A predictive score to identify hospitalized patients' risk of discharge to a post-acute care facility

Authors:

Martine Louis Simonet (martine.louissimonet@hcuge.ch)
Michel P Kossovsky (Michel.Picard-Kossovsky@hcuge.ch)
Pierre Chopard (Pierre.Chopard@hcuge.ch)
Philippe Sigaud (Philippe.Sigaud@hcuge.ch)
Thomas V Perneger (Thomas.Perneger@hcuge.ch)
Jean Michel Gaspoz (Jean-Michel.Gaspoz@hcuge.ch)

Version: 4 Date: 11 June 2008

Author's response to reviews:

Dear Editor,

thank you for providing us the opportunity to publish our manuscript in BMC Health Services Research.

As requested in your last email, we made the following additions:

The first three points from Dr Kane's review are now included in the manuscript, as part of the Discussion on the limitations of the study (pages 16, 17 and 18)

All the formatting requests have been made according to the formatting checklist.

Best Regards

Martine Louis Simonet, MD